Author’s response to reviews

Title: Interstitial cystitis antiproliferative factor (APF) as a cell-cycle modulator

Authors:

Hani H Rashid (hani_rashid@urmc.rochester.edu)
Dr Jay E Reeder (jay_reeder@urmc.rochester.edu)
Mary J O'Connell (mary_o’connell@urmc.rochester.edu)
Chen-Ou Zhang (czhan001@umaryland.edu)
Edward M Messing (edward_messing@urmc.rochester.edu)
Susan K Keay (skeay@medicine.umaryland.edu)

Version: 3 Date: 5 Apr 2004

Formatting changes and typographical error corrections were requested by the editor. All of these have been incorporated into the manuscript without exception.

A suggestion to convert Table 1 into a figure was made. The table contains 2 replicates of 6 experimental conditions and 5 population fractions. Calculation of mean values for each replicate and population fraction ignores the fact that these values are bound at 100% for each replicate. We are left with the option of a graph of 30 or 60 values and it has been our experience that graphs with this level of complexity can be difficult to interpret. The current table allows the reader to evaluate variability and the impact of treatment and dose.